Author's response to reviews

Title: Comparison of darbepoetin alfa dosed weekly (QW) vs. extended dosing schedule (EDS) in the treatment of anemia in patients receiving multicycle chemotherapy in a randomized, phase 2, open-label trial

Authors:

Lee Schwartzberg (lschwartzberg@westclinic.com)
Ronald Burkes (rburkes@mtsinai.on.ca)
Barry Mirtsching (bmirtsching@cortpa.com)
Timothy Rearden (reamdentp@aol.com)
Peter Silberstein (PeterSilberstein@creighton.edu)
Lorrin Yee (nwmsresearch@nwmedicalspecialties.com)
Amy Inamoto (ainamoto@amgen.com)
Tom Lillie (tlillie@amgen.com)

Version: 6 Date: 20 October 2010

Author's response to reviews: see over
Thank you for the additional opportunity to format and make final corrections to our manuscript. Per your instructions, the following changes have been made:

Submission system
1. Minor changes were made to the abstract, so that the abstract in the submitter system matches the abstract in the manuscript.

Manuscript
2. The “Concise abstract for TOC” was removed since this item is not required for a research article according to the BMC Cancer Instructions for Authors.
3. Minor corrections have been made throughout the text.

Thank you again for the opportunity to publish our manuscript in BMC Cancer. Please do not hesitate to contact me if any additional issues should arise with this submission.

Sincerely,

Lee S. Schwartzberg, MD, FACP
West Clinic, 100 N Humphries Blvd.
PO Box 240728
Memphis, TN 38120
Email: lschwartzberg@WESTCLINIC.com
Tel: (901) 683-0055